

# Temi di discussione

(Working Papers)

The macroeconomic effects of temperature surprise shocks

by Filippo Natoli





# Temi di discussione

(Working Papers)

The macroeconomic effects of temperature surprise shocks

by Filippo Natoli

Number 1407 - March 2023

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Lorenzo Braccini, Luca Citino, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Dario Ruzzi, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

## THE MACROECONOMIC EFFECTS OF TEMPERATURE SURPRISE SHOCKS

## by Filippo Natoli\*

### Abstract

The question of how climate change and weather fluctuations affect the economy is high on the economic research agenda, but the quantification of the effects of temperatures at infra-annual frequencies still remains an open issue. Using daily county-level data since 1970, I construct quarterly temperature shocks for the United States that capture the average surprise effect of very high and low temperatures in each county and quarter, isolating their unanticipated component. Unfavorable temperature shocks are found to reduce GDP, consumer prices and interest rates, pointing to a slowdown in aggregate demand.

## JEL Classification: C32, E32, E52, Q54.

**Keywords**: temperature shocks, business cycle, climate change, monetary policy. **DOI:** 10.32057/0.TD.2022.1407

## Contents

1. Introduction	
2. Related literature	7
3. Transmission channels	
4. Constructing the shock	
5. Data	
6. The impact of temperature shocks on the US economy	
7. Robustness exercises	
8. Other estimates	
9. Conclusions	
References	
Appendix A. Diagnostic on the shocks	49
Appendix B. Forecasting surprisingly hot and cold days	
Appendix C. Narrative on the shock	50
Appendix D. Data sources	

<sup>\*</sup> Bank of Italy, Directorate General for Economics, Statistics and Research.

## 1. Introduction<sup>1</sup>

Understanding the multifaceted effects of climate change is of utmost importance for the design of appropriate climate policies. A long standing literature explores the different implications of rising temperatures for advanced and developing countries or in hot and cold regions, focusing on the low-frequency effects on GDP growth (see Dell et al., 2014 and Carleton and Hsiang, 2016 for a review). This evidence has two potential limitations. One is that high-frequency fluctuations in temperatures may be as important as low-frequency trends, if not more (Kotz et al., 2021). The other one is that observed temperature fluctuations have both a predictable and an unpredictable component, and it is not obvious a priori that these affect economic outcomes in the same way. While a nascent literature has started to investigate temperature impacts at business cycle frequency, it has shown conflicting evidence on key outcomes such as the inflationary or deflationary effects, impairing the still under-investigated evaluation of their monetary policy implications. This lack of consensus lies on the identification of the unexpected component of temperatures, a crucial element to capture the effects of within-year fluctuations due to the existence of local trends and seasonal patterns.

I take up these issues and propose a new way to construct unexpected temperature shocks using county-level data. For this purpose, I resort to a concept that is common in the empirical macro literature, which is that of surprise shocks (see Ramey, 2016 and Nakamura and Steinsson, 2018b). Using daily county-level temperatures, I obtain quarterly US-wide shocks in two steps. First, I compute quarterly temperature surprises in each county as the number of extremely hot and cold days – relatively to the county and

<sup>&</sup>lt;sup>1</sup>An earlier version of this paper circulated under the title "Temperature surprise shocks". I thank Stephen Cecchetti and the E-axes prize committee. I also thank Max Aufhammer, James Cloyne, John Cochrane, Christian Gollier, Lars Peter Hansen, Òscar Jordà, Diego Känzig, Emi Nakamura, Brigitte Roth Tran, Glenn Rudebusch, Joseph Shapiro, Jón Steinsson, Johannes Stroebel, Soroosh Soofi Siavash, Mauricio Ulate, Reed Walker, Daniel Wilson, Luca Taschini, Piergiorgio Alessandri, Pietro Catte, Valerio Ercolani, Fabrizio Ferriani, Valerio Nispi Landi, Fabrizio Venditti and seminar participants at UC Berkeley, San Francisco Fed, European Central Bank, Banca d'Italia, E-axes webinar series, 4th Behavioral Macro Workshop, the ESCB Research Cluster on Climate Change and WTSE 2022 for their helpful comments and suggestions. I also thank Luca Missori for making textual analysis on FOMC meeting transcripts. The views expressed in this paper are those of the author and do not necessarily reflect those of the Bank of Italy. All the remaining errors are mine.

quarter of the year – in excess those agents expect at the end of the previous quarter, based on past realized temperatures. Temperature beliefs are constructed over multi-year spans reflecting the documented learning behavior regarding climate-related phenomena (Kelly et al., 2005; Deryugina, 2013; Moore, 2017, among others), and updated yearly to reflect salience of the most recent temperature realizations. Second, I average county-level surprises based on counties' weight on the national economy. By construction, the obtained series of US-wide unanticipated shocks nets out any seasonal and secular variation in the temperature distribution, including local (and potentially highly heterogeneous) trends in temperature levels and volatility. Economically, such shock reflects the surprise, at quarterly frequency, of a higher/lower number of extremes than expected, and can be used as a way to quantify the effect of severe weather on the economy, above and beyond any adapting behavior of agents to expected temperatures.

I construct temperature shocks starting from the beginning of the most recent phase of the global warming era (the 1970s), up to the end of 2019.<sup>2</sup> I then use these shocks to explore their effects on the US economy in a local projection framework and I get three main results. First, unfavorable temperature surprises have a significant negative effect on economic activity Quantitatively, a one-standard deviation positive shock (i.e., about 4 severe weather days more than expected) causes real GDP to decrease in the same quarter, with a maximum contraction of 0.3% after 2 years. Second, temperature shocks act mainly on the demand side of the economy, as the consumer price index also decreases following the shock. Moreover, investments and durable consumption fall more than non-durable expenditures, which is consistent with unexpected weather events raising awareness over future climatic risks (as in Choi et al., 2020 and Hong et al., 2020, among others) and inducing adaptation measures and precautionary behavior. In line with those impacts, short- and long-term interest rates on government bonds decrease, pointing to an expansionary monetary policy reaction. Further evidence on the behavior of the Federal Reserve, obtained by looking at the response of Greenbook forecasts to the shocks and by analyzing discussions within the FOMC, suggests that temperatures impact the Fed's short-term economic projections and stimulate some debate during official meetings. Third, the influence of temperature shocks change depending on the

 $<sup>^{2}</sup>$ The years 1970s mark the beginning of the acceleration of global warming trends in the 20th century.

season and the type of surprise, as the effects are largest in summer and, on average, when shocks originate in the left tail of the temperature distribution – surprises related to cold days.

My contribution to the literature is twofold. First, I bring into the climate debate key insights on identification developed in the macroeconomics literature. The temperature surprise shocks I construct share the economic and econometric characteristics of other popular shocks considered in the empirical macro field. Having a measure of unanticipated weather-related shock can help tackle a wider range of questions than those faced in this paper, and can be the basis of further climate research. The procedure I propose can be easily replicated for other countries, geographical aggregation levels and weatherrelated variables. Second, I quantify the aggregate short-term implications of unexpected temperature fluctuations in the United States, including the extent of monetary policy reaction. The comprehensive assessment of their economic impact made in this paper, revealing a slowdown in aggregate demand, a medium-run fall in consumer prices and a negative effect on interest rates, stands out in the literature. In this respect, looking at temperature shocks in isolation is key, as temperatures impact the economy through a set of transmission channels that can shape output, prices and interest rates differently than other weather-related extreme events.

The paper is organized as follows. Section 2 reviews the theoretical and empirical contributions related to our study. Section 3 illustrates the possible transmission mechanisms of temperature variations to the economy. Section 4 discusses the key issues surrounding the identification of an exogenous temperature shocks and describes how the shock is constructed. Section 5 presents the data used in the analysis and works out the shock series. Section 6 proposes an empirical application to estimate the domestic effect of temperatures on the US economy using the previously constructed shock. Section 7 and 8 provide robustness checks and additional findings, and Section 9 concludes.

## 2. Related literature

The effects of climate change on the economy are found to be multifaceted. Increasing temperatures substantially raise mortality rates (Carleton et al., 2022) and reduce activity and growth, especially in hot and poor countries (Dell et al., 2012; Burke et al., 2015;

Burke et al., 2018; Acevedo et al., 2020; Kiley, 2021; see Dell et al., 2014 and Carleton and Hsiang, 2016 for a review of the literature). Advanced economies appear not to be immune either, as more recent analysis points to a negative impact of temperatures even in most G7 countries (Berg et al., 2021) and OECD economies (Ciccarelli and Marotta, 2021). Evidence of an impact of temperatures in the United States, initially mixed, is also growing. Indeed, US output is negatively affected in a wide range of industries (Hsiang et al., 2017; Colacito et al., 2019), with income per capita losses being concentrated during business days (Deryugina and Hsiang, 2014).<sup>3</sup> From a wider perspective, extreme weather events are found to have increased their negative impact on the US economy in the most recent decades, according to Kim et al. (2021). However, the propagation of the effects of temperatures throughout the economy and their implications for consumer prices still remain an open issue, especially in the case of the United States. For example, severe weather (including extreme temperatures) are found to be, in the medium run, either inflationary (Kim et al., 2021 and Makkonen et al., 2021) or deflationary (Faccia et al., 2021), with radically different implications for the conduct of monetary policy. Moreover, challenges in isolating the economic effects of temperatures remain, also because their implications differ substantially in different seasons and sectors (Addoum et al., 2021).<sup>4</sup> I contribute to this literature by shedding light on the impact of temperatures shocks on the US economy, exploring their effects beyond aggregate output including consumption, investment, consumer prices and bond yields.

From a methodological point of view, this paper contributes to the recent strand of the literature that aims at quantifying the short-term impacts of climate change, notably using temperature fluctuations. A common empirical approach is to retrieve shocks in a recursively-identified structural VAR, with average temperatures included as the most exogenous variable (Donadelli et al., 2017); the same logic has been applied to extrapo-

<sup>&</sup>lt;sup>3</sup>A significantly negative effect is also found on agriculture (Fisher et al., 2012; Burke et al., 2015), despite earlier evidence suggested no impact (Deschênes and Greenstone, 2007).

<sup>&</sup>lt;sup>4</sup>While pointing to a detrimental effect on firm activity, micro evidence is still mixed. Some papers do not find any consequence of high temperatures on firm sales and labor productivity (Addoum et al., 2020), while others document negative effects on upstream firms in the supply chain (Custódio et al., 2021; Pankratz and Schiller, 2021) and a reduction in the number of employees and firm establishments in the medium run (Jin et al., 2021).

late shocks from other weather series.<sup>5</sup> Other studies employ the so-called temperature anomalies, constructed by re-scaling actual temperatures with averages of a pre-global-warming (or pre-1970) reference period, and use them as a direct measure of exogenous temperature variations (Makkonen et al., 2021, among others); finally, other works adopt a down-scaled version of the panel framework used in the multi-country literature to analyze the effect of average temperatures along the business cycle. I document potential identification issues that arise by applying standard methods to retrieve shocks from temperature variations, and propose a way to tackle them by using daily data and a granular geographic coverage.<sup>6</sup> In particular, I propose a method to compute shocks that capture, from a business cycle perspective, the unexpected component embedded in daily temperature realizations. To take into account the fact that temperature events in different areas and seasons can produce profoundly different aggregate effects, my application to the US case is based on county-level data – as it is done by Moscona and Sastry (2022) to study the technological response to climate change – and takes into account seasonal patterns as in Colacito et al. (2019) and Addoum et al. (2021).<sup>7</sup>

My notion of temperature shock is based on the variation over time in the shape of the distribution of local temperatures. In this respect, this paper connects to those studying the economic implications of actual or expected temperature volatility (Kotz et al., 2021; Donadelli et al., 2021; Alessandri and Mumtaz, 2021; Diebold and Rudebusch, 2022; Bortolan et al., 2022). While these papers isolate changes in the second moment of the temperature distribution, my shock is defined over the number of severe temperature days experienced by agents, linking more closely to the thickness of the distribution's tails.

The idea of constructing macroeconomic temperature surprises is inherited from the empirical macro literature, in which the identification of shocks typically rely on extrapolating an exogenous component from policy announcements or decisions (see Nakamura

<sup>&</sup>lt;sup>5</sup>Gallic and Vermandel (2020) uses a de-trended version of the soil moisture index, a measure capturing the combined effect of temperatures and precipitations, in a VAR model; Cashin et al. (2017) includes deviation of the Southern Oscillation Index (SOI) from their historical averages in a Global VAR framework.

<sup>&</sup>lt;sup>6</sup>Concerns regarding the commonly used methods to estimate temperature effects – particularly with panel data – are raised, for different reasons, by Berg et al. (2021).

<sup>&</sup>lt;sup>7</sup>My analysis is focused on identifying economic impacts of weather variations. For a review of the literature investigating the link with longer run climatic effects, see Kolstad and Moore (2020).

and Steinsson, 2018b for a review of the literature). For example, the notion of surprise shocks related to monetary policy refers to the contemporaneous surprise component of monetary policy announcements (Gurkaynak et al., 2005; Gertler and Karadi, 2015; Nakamura and Steinsson, 2018a; Miranda-Agrippino and Ricco, 2021 among others); regarding fiscal policy, government spending surprises can reflect the unanticipated component of public spending decisions (Forni and Gambetti, 2016, among others). As in the case of policy surprises, what matters here is defining agents' ex-ante temperature expectations in the correct way: in a quarterly setting, what agents expect in a given year can be well proxied by temperatures directly experienced during the same period in the most recent years, i.e. those agents are used to. Defining expectations based on past temperatures connect to the literature on learning from climatic events (Kelly et al., 2005; Deryugina, 2013; Moore, 2017; Kala, 2019; Choi et al., 2020; Pankratz and Schiller, 2021) and, more generally, to learning from direct experience (Malmendier and Nagel, 2015).

Last, my paper links to those investigating the effects of climate change on consumer prices and the reaction of monetary policy. With respect to output effects, the implications for prices are less clear in the literature, sometimes providing opposite evidence (Mukherjee and Ouattara, 2021; Faccia et al., 2021). This can be due to the fact that temperatures might have relevant demand-side – other than supply-side – effects, which might offset the final impact on consumer prices at some particular horizons. I find that the effects of temperatures on prices are skewed towards a price fall in the medium run. This price response is broadly in line with that found in Faccia et al. (2021) for a panel of countries, for which the impact is initially positive and it becomes negative in the longer term. On monetary policy, the literature has mainly focused on how changes in the monetary stance might have implications for climate and how central banks might cooperate to stimulate the low-carbon transition (Hansen, 2021, among others), almost disregarding the effects of actual weather occurrences on its conduct. I fill this gap by documenting that monetary policy responds to the economic damage caused by temperature shocks by cutting short-term rates, with effects passing through the whole term structure of government bond yields.

#### 3. Transmission channels

The economic effects of temperatures may unfold through different transmission mechanisms. The one that received most attention in the literature is the physical impact of extreme temperatures on human health, the so called heat stress channel. As extreme temperatures have the potential to cause several illnesses (e.g., heat strokes), they can hit labor supply by shrinking hours worked and inducing a fall in individual productivity in temperature-exposed working tasks (Cachon et al., 2012; Somanathan et al., 2021).

However, temperatures might also hit the economy from the demand side. Indeed, some papers have documented significant behavioral effects, according to which extreme temperatures would discourage open air activities: perception of waiting time worsens in hot days (Baker and Cameron, 1996) and social interactions with strangers are felt as more unpleasant (Griffit and Veitch, 1971), reducing time allocated to outdoor leisure (Graff Zivin and Neidell, 2014). These effects can put downward pressure on consumer spending, for example through a decrease in shop retail sales (Starr-McCluer, 2000; Roth Tran, 2022). Another demand-side effect works through an expenditure channel. Upward trending temperatures will raise electricity demand (McFarland, 2015), increasing energy expenditures for households and firms. For the latter, the need of cooling/heating down work spaces or production processes in times of exceptional highs and lows might entail unanticipated expenses, reducing the available liquidity and, possibly, eroding profits. A third channel of transmission, which also works through the demand side, is related to the uncertainty over future climatic developments. Temperature oscillations, if wide or frequent enough, can raise attention towards the future repercussions of climate change – wake-up call effect – thereby influencing decision making: this is documented in Choi et al. (2020), who link extreme temperature episodes to larger financial investments in green than brown assets. As in the case of financial investors, temperatures can induce preference shifts among households and firms, who can modify their attitude to hedge against the future consequences of climate change. For example, forward-looking entrepreneurs might undertake adaptation investments to make their business more resilient to temperatures, which can be detrimental for short-run firm performance if this crowds out other productive investments.<sup>8</sup>

<sup>&</sup>lt;sup>8</sup>In the longer run, the overall effects of temperatures might be heterogeneous across firms depending

All in all, the economic effects of temperature oscillations can be diverse, potentially impacting both the consumption and investment components of output. In what follows, I explain how my shock is constructed, starting from methodological issues and insights, and test its effects on GDP, consumer prices and interest rates.

#### 4. Constructing the shock

## 4.1. Key points

Using temperatures to quantify the economic effect of weather variations has some clear advantages with respect to other weather events: temperatures are continuously recorded and collected simultaneously across the country, allowing to compute highfrequency statistics with granular geographic detail. However, constructing shocks using temperaturespresents non-trivial challenges: temperatures are a very local phenomenon, they have strong seasonal components and are pretty unforecastable beyond very short time spans.

In order to quantify their economic impact at infra-annual frequencies, the literature has so far adopted one of the following empirical strategy: (1) estimating country-level shocks in a time-series framework, where shocks are retrieved within VAR models based on fluctuations in country-averaged temperatures levels (or temperature anomalies); (2) estimating local impacts using fixed-effect panels at sub-national level, where local shocks are identified as deviations of average temperatures – or the count of days with temperatures beyond certain thresholds – from their long-run means. Both of these methods rest on some problematic assumptions. For example, average temperatures are not suitable to construct "shocks" based only on their positive/negative time variations, as those can be either good or bad for the economy depending, for example, on temperature levels – passing from 50°F to 60°F and from 85°F to 95°F can have even opposite economic effects. As what can be labeled as severe weather varies across location and over time, using statistics based on the occurrence of temperatures beyond fixed thresholds (eg., 85°F in all locations and seasons) is also problematic. A different issue related to the use of anomalies or fixed effects is that deviations of temperatures vis-à-vis long-run averages

on whether adaptation measures spur innovation and technological progress, as explored in Cascarano et al. (2022).

may not be unexpected, as agents can adapt over time to higher and more volatile temperatures and update their beliefs accordingly. Last, when dealing with the country-level dimension, shocks built on country-averaged temperatures may not properly reflect the aggregation of local shocks, as the latter may depend non-linearly from local temperature variations.

To overcome these issues, macroeconomic shocks based on temperatures should be defined at the most granular geographic dimension as possible, to identify their economic impact on top of any expectations about the incidence of "good" and "bad" temperatures in each localities and periods. As stated in Ramey (2016), a shock should represent "either unanticipated movements in exogenous variables or some news about future movements in the exogenous variable". Finally, identified temporary local shocks need to be compounded to get an aggregated macro shock.

I use daily temperatures in each US county to construct a nation-wide shock at the quarterly frequency. The shock design is based on the following arguments. First, as science suggests, temperatures turning very hot or cold in a short period of time can be considered as exogenous to current and recent past economic activity, as feedback effects to local temperatures from human-generated CO2 emissions unravel only in the longer term. Second, very high ad low temperatures are intrinsically undesirable and, as documented in the literature, generate mostly detrimental effects on the economy. Indeed, while the direction of the impact can be different by season and depend on the type of business exposed, studies recognize that, overall, both extremes are bad for the economy. At the hearth of this outcome lies, for example, the U-shaped relationship between temperatures and mortality documented for the United States by Deschenes (2014) and Barreca et al. (2016), among others.<sup>9</sup> Third, I argue that while agents can be able to find a workaround to isolated extreme temperature episodes, e.g., by rescheduling working tasks or outdoor activities, exposure to very hot or cold temperatures could become impossible to avoid if these events are frequent enough within a short time span, with negative effects on human and firm activity accumulating over time. This might be so because there are limits to adaptation in the short run, and time constraints (notably

<sup>&</sup>lt;sup>9</sup>A similar U-shaped relationship is found with respect to the impact on crop yields, see Schlenker and Roberts (2009).

in business) that impede to postpone any important activity.<sup>10</sup>

All these arguments suggest that a shock to temperatures can be inferred by looking at the incidence of multiple extreme temperature episodes within a specific period of time. I choose to work at quarterly frequency for different reasons. One reason is that quarterly performance matters in firm's business – e.g., publicly-traded companies file reports quarterly – so substitution of work activities over time, for example due to unfavorable temperatures, might be less feasible across quarters. Another reason, from an empirical point of view, is that quarterly frequency is convenient to explore the effects of temperature shocks on official GDP figures. Last, and most important, quarters roughly coincide with calendar seasons, which are quite homogeneous in terms of temperatures. In this respect, it is important to note that temperature highs or lows in each seasons can be damaging even if they do not reach extreme levels in absolute terms. For example, the impact of an exceptionally cold summer, while not reaching winter lows, can be nonetheless material for tourism. If this is so, to be surprising extreme realizations need to be evaluated with respect to what the distribution of temperatures is expected to be for that period.

In this perspective, as the shape of the entire temperature distribution changes over time, a proxy of agents' beliefs on current temperatures must be based on past realizations that cannot go too far back in time. One reason for that, from an economic viewpoint, is that agents have memories and learn from their past experience: this form of experiencebased Bayesian learning, which takes multiple years, has been documented to drive the dynamics of beliefs also related to global warming (Kelly et al., 2005; Deryugina, 2013; Moore, 2017; Kala, 2019; Choi et al., 2020; Pankratz and Schiller, 2021). In my shock computation, I assume that the reference distribution for each quarter rolls over time, in order to compare current values with updated temperature beliefs. This identification procedure marks a stark difference with all past approaches based on temperature anomalies or on deviations from historical averages, which implicitly assume that agents anchor their beliefs to some average values and do not update them over time. According

<sup>&</sup>lt;sup>10</sup>For households, although available at very short horizons, daily weather forecasts may in principle help to cope with undesirable temperatures: however, evidence shows that an anticipatory behavior of future temperatures – inducing protective actions – is limited at best (Morss et al., 2010; Graff Zivin and Neidell, 2014).

to those metrics, observing a larger number of extreme realizations than in the pre-1970 period would always imply a positive shock, even though agents can be, in fact, used to them: therefore, outdated beliefs may lead to overestimate the size of the most recent shock. In the following section I provide the formula to recover my shock at local level, as well as the procedure to aggregate it to country level.

#### 4.2. A US-wide temperature shock

According to a very general framework, country-level shocks at time t can be thought as the weighted average of county-level temperature expectation errors

$$\sum_{i=1}^{k} w_{t}^{i} \underbrace{\left[ f(T_{t}^{i}) - E_{t-1}f(T_{t}^{i}) \right]}_{\text{expectation error in county } i}$$
(1)

where  $f(T_t^i)$  is the value of a temperature statistics for county i = 1, ..., k and quarter t;  $E_{t-1}f(T_t^i)$  is the expectation about  $f(T_t^i)$  made at the end of the previous quarter;  $w_t^i$ are the weights to aggregate local errors to a country-level shock. In order to define  $f(T_t^l)$  in the case of the United States, I first collect average temperatures in each county at daily frequency. For each series, I group observations by quarter and compare the within-quarter distribution of temperatures to a reference distribution, which is made by pooling daily observations recorded in the same quarter of the past years: this reference distribution is the one over which  $E_{t-1}f(T_t^i)$  are constructed. As five years is a sufficiently long period for agents to learn about the shape of the underlying temperature distribution, I construct the reference distribution based on that time span.<sup>11</sup> The reference distribution rolls over time, i.e. it is updated every year for each quarter. In each period, I compute the 10th and 90th percentiles, which are taken as upper and lower thresholds for current temperatures, i.e. the values beyond which actual observations are labeled as very high or low. In order to be perfectly aligned with expectations, I posit that the share of extreme days in current quarter must be the same of that distribution: differently, a larger number of extremes represent a positive surprise, while a lower number makes a negative one.

<sup>&</sup>lt;sup>11</sup>Pankratz and Schiller (2021) test learning periods of five, ten, and fifteen years length. As a robustness check, I construct an alternative version of the shock using a 10-year learning window as in Choi et al. (2020), see Section 8.

In formulas, let  $T_{d,q,y}^i$  being average daily temperatures in day d, quarter q and year y, recorded in county i, for i = 1, ..., K counties. Denote with  $F_{q,y}^i = \{T_{d,q,y-j}^i, j = 1, ..., 5\}$  the empirical cumulative distribution function of the reference temperature distribution for quarter q and year y, totalling  $N_{q,y}$  days. The reference temperature values are

$$\operatorname{ut}(q)_{y}^{i} = F_{q,y}^{-1}(0.9) \quad \forall \quad q = \{q1, q2, q3, q4\}$$
 (2)

$$lt(q)_{y}^{i} = F_{q,y}^{-1}(0.1) \qquad \forall \quad q = \{q1, q2, q3, q4\}$$
(3)

where  $\operatorname{ut}(q)_y^i$  and  $\operatorname{lt}(q)_y^i$  are yearly series for upper (lower) thresholds for county *i*, quarter q and year y. By combining threshold values together by quarter, I get quarterly series of upper and lower thresholds for each county, i.e.

$$ut_t^i = \{ut(q)_y^i, q = q1, \dots, q4\}$$
$$lt_t^i = \{lt(q)_y^i, q = q1, \dots, q4\}$$

The same can be done with the size of the reference distribution, yielding  $N_t = \{N(q)_y, q = q1, ..., q4\}$ . Note that I change notation for the quarter (from q to t) as q denotes quarters in yearly series, while t indicates quarterly frequency. Denoting with  $n_t$  the number of days in quarter t, county surprises are then evaluated as the number of beyond-threshold days in the current quarter in excess of those in the reference distribution. Provided that, by construction, the share of extreme days in the reference distribution is 20 percent, and that this number must be rescaled to a single quarter dimension, county surprises can be expressed as

$$\operatorname{county\_surprise}_{t}^{i} = \underbrace{\sum_{d=1}^{n_{t}} \left[ I(T_{d,t}^{i} < \operatorname{lt}_{t}^{i}) + I(T_{d,t}^{i} > \operatorname{ut}_{t}^{i}) \right]}_{f(\operatorname{T}_{t}^{i})} - \underbrace{N_{t} \times 0.2 \times 0.2}_{E_{t-1}f(\operatorname{T}_{t}^{i})} \qquad i \in k \quad (4)$$

where I(x) is an indicator function that values 1 if x is true, 0 otherwise. County surprises are the difference between the number of hot and cold days in quarter t and the number of "extremes" in the reference distribution for that quarter. The underlying idea is that the reference distribution, which updates quarterly, represents the information set of economic agents, who directly experienced a range of temperatures for that season in the past. As they have no reason to foresee any significant change in the distribution with respect to the very recent past, agents expect that the same number of hot and cold days also occur in current quarter. In this setting, temperatures go beyond expectations if the number of extreme days exceed (or is below) what agents expect.

In order to make a US-wide surprise shock, I make a weighted average of county-level surprises occurred in the k counties by quarter:

US\_shock<sub>t,y</sub> = 
$$\sum_{i=1}^{K} \left( \text{county}\_\text{surprise}_{t}^{i} \times w_{y-1}^{i} \right)$$
 (5)

where w are county-level weights, proxying counties' vulnerability to temperatures, which vary at annual frequency. Weights are lagged to capture the ex-ante exposure to temperatures.

## 4.3. Insights

The constructed shock presents the following characteristics. First, it is designed to have an unambiguous economic effect: a positive shock, meaning an unexpected increase in the occurrence of severe daily temperatures, should impact the economy in one direction (eg., weigh negative on output), while a negative one – milder temperatures than expected – in the opposite way. Second, it is season-specific, as surprises are measured within the same season. Therefore, the largest economic effects during the year do not necessarily come in winter and summer when temperatures show record lows and highs, as what matters is how temperatures deviate from their seasonal norm. Third, as the shock is measured with respect to updated expectations, it nets out longer-run phenomena such as the intensification of climatic trends – upward-drifting and increasing volatile temperatures – and adaptation to extreme temperatures: agents learning about the evolution of temperatures can continuously adjust their resilience to them. Last, the shock is two-sided and surprises coming from extremely hot and cold days are treated as equivalent. This can be considered as a neutral assumption, as taking a stance on which tails matter more in each season is not straightforward (Addoum et al., 2021).<sup>12</sup>

Overall, the shock measures the size of the variation in the distribution of temperatures in the short run. Being an aggregation of county-level surprises, it will depend more on counties that have a higher weight in the US economy. As explained in Section

<sup>&</sup>lt;sup>12</sup>In a robustness exercise, I decompose the US-wide shock into heat and cold shocks and explore their effects separately by season using local projections, see Section 8.

5.1, I follow the literature and propose alternative methods to rank counties based on the potential economic damage due to unfavorable temperatures, and use them to weight county-level surprises to get the US-wide shock.

### 5. Data

### 5.1. Temperature data

To construct my shock, I rely on two data sources. The first one is the gridded air temperature database for continental US taken from the Northeast Regional Climate Center. From that source I extract daily temperatures, i.e. the mean value of temperatures recorded during the 24 hours, averaged at county level. I consider data starting from Jan 1, 1970, i.e. when the human-induced global warming trend started to reinforce, and ending on Dec 31, 2019. To construct the weights to aggregate county-level surprises, I take annual data on US counties' economies from the Census Bureau, from 1969 to 2018. In the baseline formulation of the shock, I consider county-level population and construct weights as the county share over nation-wide values. The rationale for this choice is that the higher the population, the higher the incidence of agents that are exposed to extreme temperatures, so the higher the potential impact on human health and the economy. As a robustness, I also consider land extension – used as an alternative to population in Colacito et al. (2019) – and other weighting schemes such as the number of employed people, personal income or county GDP to weight temperature surprises: while population, employed people and land extension also reflect temperature exposure, personal income and county-level GDP proxy counties' economic weight in the US economy, independently from their exposure. Merging the two datasets yields time series of temperatures and weights for 3053 counties. This sample is highly representative of the US economy, covering more than 98% of the country's population, jobs and personal income as of 2019.

#### 5.2. County-level statistics

Figure 1 represents county-level statistics over geographic maps of the United States. Panel 1 shows the growth rate of temperatures in the years 2010s with respect to the 1960s. On average, temperature grew by 1.7%, but climatic trends have been quite diverse, with some counties experiencing temperatures rising by more than 8%, while



Panel (1): Growth rate of temperatures, 2010s vs 1960s averages (percent)

Notes: The figure shows county-level statistics of temperatures, surprises and population. Panel (1) shows the historical growth rate of temperatures in the most recent phase of the global warming era (post-1970), computed by comparing average temperatures in the years 2010s with average temperatures in the 1960 decade. Panel (2) shows the time series variance of county-level surprises, a measure of their average historical size, computed for the period 1975q1–2019q4. Panel (3) displays population levels in 2019.

others in which temperatures decreased by 5%. Overall, northeastern counties and the west coast experienced the steepest temperature increase. The picture looks different when evaluating the size of county-level surprises occurred over time, which depend on the

evolution of the entire temperature distribution: surprise shocks in northeastern counties have been small on average, while they have been very large in the south (Panel 2). Overall, what matters is the combination of shocks and weights: even small surprises can be important if occurred in highly exposed (or rich) counties. Panel 3 shows counties' population in 2019. At that time, the 20 most populated counties, covering about 20 percent of US population, were mostly in California, Texas, New York and Florida.

#### 5.3. The shock series

The time series of the US-wide shock is displayed in Figure 2. The shock is expressed as the total number of surprisingly hot and cold days per quarter, and reaches a minimum at -9 and a maximum at 14. The shock embeds the basic econometric requirements to be used in empirical analysis: it is zero-mean and serially uncorrelated (see Appendix Appendix A). From an economic point of view, it features the three characteristics, explained in Ramey (2016), which make it suitable for macroeconomic applications: it is exogenous with respect to current and lagged outcome variables, it can be considered as uncorrelated with other exogenous economic shocks and it represents unanticipated movements in an exogenous variable.

Figure 2: US-wide temperature shock



Looking at the plotted series, one thing that catches the eye is that temperature shocks have not been particularly large in the last 15 years with respect to the earlier period. Indeed, the volatility of the shock series computed on a rolling 10-year window slightly decreases throughout the sample, from a peak of 5 to about 3.5 days. It suggests that adjustments in the shape of the temperature distribution have been largest – inducing greater surprises – in the early part of the sample than in recent times. This evidence reverses the common wisdom that climate change is generating shocks of increasing size. In fact, my shock dynamics is not in contrast with the intensification of climate-related weather events (including temperatures extremes), as climate indicators such as the Actuaries Climate Index show.<sup>13</sup> As I take out predictable climatic trends to capture the unexpected component of temperature fluctuations, it is well possible that large weather episodes with respect to the pre-1970 period have become increasingly frequent, but also less surprising than in the past.<sup>14</sup>

As explained, shocks are computed with respect to temperature expectations that are backward-looking and based on the closest, past temperature data. These features are tested in two separate robustness exercises (see Appendix Appendix B and Section 7.1). Regarding the first one, results suggest that publicly available temperature forecasts cannot be of help in forming expectations for the quarter because days that are labeled as surprisingly hot or cold under this procedure are also more difficult to predict at oneand two-day horizons, as an exercise using Washington DC data shows. The second exercise reveals that, if expectations are modeled as anchored at some pre-sample level (eg., the average of temperatures in 1970-1974 in the same quarter), the incidence of positive shocks – and the role of heat over cold surprises – are increasingly overestimated over time.

The peaks in Figure 2, i.e. largest positive values of the shock, are recorded in 1977q1, 1989q4, 2000q4 and 2003q1. For a description of the main events surrounding those dates, see Appendix Appendix C. All of them were mainly due to abnormally cold periods. However, heat-related surprises have also been frequent in the history of US temperatures. To have a flavor on how both extremes contribute in shaping US-wide shocks, I inspect the incidence of surprisingly hot and cold days in the shock series within

<sup>&</sup>lt;sup>13</sup>The Actuaries Climate Index is a composite indicator of the frequency of a set of climate-related natural events including extreme temperatures, precipitation, wind, drought and sea level rise (American Academy of Actuaries, Canadian Institute of Actuaries, Casualty Actuarial Society and Society of Actuaries, 2020). The components are constructed as anomalies, i.e. in difference with respect to a fixed reference period in the past (1961-1990).

<sup>&</sup>lt;sup>14</sup>This higher predictability speaks to the evidence of an upward trend in annual temperature volatility found for a panel of countries in Alessandri and Mumtaz (2021), over which the US economy may have adapted over time.

each quarter. Figure 3 displays positive US shocks in a bar plot where observations are labeled as orange or blue depending on whether surprisingly hot or cold days prevailed at each point in time. Abnormally hot and cold quarters have been almost equally frequent, with the former being slightly more (55% of times). Moreover, their incidence has been mostly balanced throughout the sample, with no apparent clusters in the most recent period. This finding also dispels the myth that, in a global warming era, abnormally hot days largely predominate over cold ones. In Section 8, I break down the US-wide shock into heat and cold shocks and estimate the economic effect of each component separately, finding comparable albeit not equal effects.



Figure 3: Hot vs. cold quarters

Notes: Positive shocks only. Orange (blue) bars: quarters in which surprisingly hot (cold) days predominate.

#### 6. The impact of temperature shocks on the US economy

In this section I use the constructed shock to evaluate the effect of temperature surprises on the US economy. In the following, I describe my approach and comment on the main findings.

#### 6.1. Impulse-response analysis

I estimate the response of US domestic variables to the previously constructed shock using the local projections framework of Jorda (2005). Impulse response functions (IRFs) are obtained from the following linear regressions:

$$y_{t+s} = \alpha_s + \beta_s \text{ US\_shock}_t + \psi_s(L)\mathbf{X}_{t-1} + u_{t+s} \qquad s = 0, 1, 2, \dots, H$$
(6)

where t are quarters, y is the target variable and  $\mathbf{X}_t$  is a vector of controls.<sup>15</sup> Estimates are made separately for each time horizon s and for each dependent variable. IRFs are defined by the sequence  $\{\beta_s\}_{s=0}^H$ , and inference is performed with Newey-West standard errors.

#### 6.2. Target variables

In my baseline estimates, I test the effects of temperature shocks on the following set of domestic dependent variables: GDP, private consumption and investment (all in real terms); the CPI index, 3-month interest rate and the 10-year Treasury yield.<sup>16</sup> On the right hand side of Equation 6, I include a tight set of controls: linear, quadratic and cubic time trends, seasonal dummies, eight lags (up to 2 years) of the shock and of the aforementioned variables, and the 1-year less 3-month yield to control for the impact on the short-term portion of the yield curve.<sup>17</sup> A summary of all data used in this paper is reported in Appendix Appendix D. Impulse responses are estimated on a 16-quarter horizon, and displayed with 68% and 90% confidence bands. The baseline estimates are carried out on the full sample, going from 1975 Q1 to 2019 Q4.

## 6.3. Results

Impulse response functions from a one-standard deviation shock are displayed in Figure 4. In response to a positive (unfavorable) temperature surprise, real GDP in the United States significantly declines on impact, with the effect becoming larger over time reaching a trough between 1 and 2 years after the shock. Both private consumption and investment shrink, with investment being much more impacted – response is five times larger at the trough.<sup>18</sup> With some lag, the CPI index also decreases after the shock,

 $<sup>^{15}</sup>$ Note that, with respect to previous notation, I here suppress subscript y to indicate years.

 $<sup>^{16}\</sup>mathrm{All}$  variables except the 3-month rate and the 10-year yield are expressed in natural logarithm.

<sup>&</sup>lt;sup>17</sup>In the set of controls, variable lags are reduced to 4 in estimates made on shorter samples.

<sup>&</sup>lt;sup>18</sup>The latter also sees some rebound, possibly due to an increase in climate adaptation investments, which is not enough to significantly raise output in the longer term.

suggesting that demand-side effects dominate over supply-side ones. Coherently with the slowdown in both output and prices, short-term interest rates also decline, suggesting an expansionary monetary policy response. This effect passes through the long end of the government yield curve, as the 10-year yield also falls significantly.

In terms of size, the economic effects can be quantified as follows: a positive shock in a single quarter by one standard deviation, which is equal to 4.3 days (5% of the days in the quarter), entails a decrease in real GDP by 0.1% on impact, which increases up to -0.3% in about two years, when also lagged effects come into play.<sup>19</sup> Temperature effects growing over time speak to the findings in Lemoine (2021), showing that weather shocks induce different forms of adaptation adding up dynamically to the direct effects of the shocks. However, as temperatures are found to eventually reduce consumer prices, there should also be other explanations for such persistent effect, pointing to a drag on aggregate demand. Looking beyond GDP effects, the shock also implies an overall decrease in real private consumption by 0.2%, in real investment by 1.2% and in the CPI index by 0.2%. Regarding interest rates, it generates a fall in the 3-month rate by 35 basis points, and in the 10-year yield by 18 bps. Taken together, the effects of GDP, CPI and interest rates stand out in the literature. In particular, this is the first evidence, as far as I know, of a significant response of US interest rates to temperature fluctuations.

<sup>&</sup>lt;sup>19</sup>A negative impact of temperature shocks on GDP is in line with the negative effects of an increase in average temperatures found in Deryugina and Hsiang (2014), Hsiang et al. (2017) and Colacito et al. (2019).





Notes: 68% and 90% confidence bands.

#### 6.4. Shock transmission

In this section I explore the impact of the shock to a number of additional macroeconomic variables to help figure out how temperature surprises propagate throughout the economy, with a focus on demand components. Figures 5 and 6 display the breakdown of the response of private consumption and investment. Figure 5, on consumption, shows that the effect of the shocks are stronger on the consumption of durables than on that of non-durables or services. Together with the evidence of investment being more impacted than consumption (Figure 4), this suggests that temperature shocks can weight on longterm economic beliefs, maybe inducing a heightened risk aversion or higher economic uncertainty due to weather- and climate-related risks. On the other hand, the impact on non-durables expenditures can also be associated to the effect of weather on retail sales, as in Roth Tran (2022). Among non-durables, the figure also singles out the effects on energy consumption, which increases on impact (driven by a rise in the amount of electricity consumed for air conditioning or heating purposes) and becomes negative in the medium run, indicating that low aggregate demand may also end up depressing that for energy. Figure 6, on investment, shows that investment in equipment, both residential and non-residential, are the first to be cut to cope with the surprising temperature events,

while other components, including R&D investment, are reduced with some lag.<sup>20</sup>



Figure 5: Consumption breakdown

#### Figure 6: Investment breakdown



Figure 7 investigates possible differences in the shock transmission across sectors. As official figures on quarterly output by economic industry are only available since 2005, I

<sup>&</sup>lt;sup>20</sup>On the contrary, investments in non-residential structures increase on impact, possibly because of some short-term relocation as in Castro-Vincenzi (2022).

here display the impact on the real value added of the business sector for the farm and nonfarm sub-sectors (Panel a), and on GDP-deflated industrial production of manufacturing and utility (Panel b). As expected, while the effect on the farm sector is way larger – confirming that agriculture is the most exposed activity to climate change – value added in the non-farm sector also significantly declines, suggesting that temperature effects are widespread across the US economy. Within the industrial sectors, the negative effects come from a slowdown in manufacturing while utility production goes up on impact, in line with the finding on energy consumption in Figure 5. Regarding prices, the negative effects on CPI are clearly driven by non-farm sectors only, as the price index on farm value added goes up following a temperature shock, as expected.



Figure 7: Sector breakdown

Price effects are investigated in greater detail in Figure 8, where the response of the CPI is decomposed into its core (left) and volatile components: food and beverages (center picture) and energy (right picture). The price response appears overall muted on impact, with food prices responding positively – albeit non-significantly. However, after four quarters the response of both core and energy components become negative, suggesting once more that temperatures act mainly by lowering aggregate demand.

Finally, figure 9 displays labor market effects, focusing on employment per capita (employment/population), hours worked per worker of the business sector (total hours/employment), and labor productivity of the business sector. Following a temperature shock, employment per capita falls, remaining below the baseline for eight quarters; together with that, the number of hours worked per worker also decreases, suggesting that the effects on total labor supply come from both the extensive and the intensive margin. In addition, labor productivity falls significantly. Taken together, results for labor market variables confirm the relevance of the heat stress channel found in previous contributions.

· · · · · · · · · · · · · · · · · · ·	Figure	8:	Price	effects
---------------------------------------	--------	----	-------	---------



#### Figure 9: Labor market



## 7. Robustness exercises

In this Section I test a number of alternative assumptions to construct my temperature shock and perform robustness exercises to confirm my baseline findings.

#### 7.1. Fixed pre-sample temperature expectations

As pointed out in Section 4, it is well documented by the literature that agents learn about temperatures over time and update their temperature beliefs accordingly. This feature is embedded in the constructed shock by making the thresholds for county-level surprises (lt and ut in Equation 4) vary over time. We here evaluate how results change whether these thresholds are instead made time-invariant, i.e. anchoring expectations of the temperature distribution to some pre-sample level without updating them over time.<sup>21</sup> In formula, this alternative version of county-level surprises looks as follows

$$\operatorname{county\_surprise}_{t}^{i} = \sum_{j=1}^{n_{t}} \left[ \mathrm{I}(T_{d,t}^{i} < \bar{\mathrm{lt}}^{i}) + \mathrm{I}(T_{d,t}^{i} > \bar{\mathrm{ut}}^{i}) \right] - N_{t} \times 0.2 \times 0.2 \qquad i \in k \quad (7)$$

where  $\bar{lt}^i$  and  $\bar{ut}^i$  are 10th and 90th percentiles computed by quarter over the years 1970-1974. County-level surprises constructed in this way are then used to made an alternative version of the US shock according to Equation 5. Figure 10 displays this alternative version together with the baseline shock series. Anchoring expectations at pre-sample level yields a distorted picture of agents beliefs, as the shock constructed in this way increasingly overestimates the incidence and size of positive shocks over time. A closer look at the composition of hot and cold surprises within positive shocks, shown in Figure 11, also reveals that the incidence of hot surprising days is increasingly overestimated, reflecting the upward temperature trend that time-invariant expectations are unable to net out.

#### 7.2. Surprise thresholds

The temperature surprise shock is constructed in each quarter with respect to some threshold values, which are the 10th and 90th percentiles of the reference distribution. I propose alternative specifications of the shock in which I modify the definition of surprising days by setting threshold values at the 5h and 95th percentiles, or at the 25th and 75th percentiles of the reference distribution. Figure 12 shows the median responses for my 6 variables of interest in these two cases, together with the baseline IRFs. Impulse responses of the two alternative specifications look very similar to the baseline, with shocks

<sup>&</sup>lt;sup>21</sup>Such alternative formulation is in the spirit of using temperature anomalies to evaluate the economic effects of current temperatures.





Notes: Red: shock based on fixed 1970-1974 thresholds. Black: baseline shock

Figure 11: Heat vs cold surprises in the shock with fixed pre-sample temperature expectations



Notes: Positive shocks only. Orange (blue) bars: hot >(<) cold surprises.

based on temperatures in the farthest part of the tails (5th-95th perentiles thresholds) having a slightly stronger impact on GDP. Correlations between baseline US shock and the two alternatives are about 96%.

## 7.3. Weighting scheme

I here compute alternative formulation of the US-wide shock using different weights to aggregate county-level surprises. In particular, instead of counties' population shares, I use the counties' share of the following variables: (1) land extension; (2) employment; (3) personal income; (4) real GDP in 2001.<sup>22</sup> The median impulse responses obtained

 $<sup>^{22}</sup>$ As county-level real GDP is only available since 2001, I made time-invariant GDP weights based on that first observation. Differently, variables (1) to (3) are used to compute annual time-varying weights as in the baseline computation.



Figure 12: IRFs to shock computed with different thresholds

Notes: 68% and 90% confidence bands.

from these alternative shocks are displayed in Figure 13. Using employment shares, personal income shares and GDP shares leave results mostly unchanged with respect to the baseline. Using land instead of population weights change responses a bit more, although remaining broadly in line with baseline results.

Figure 13: IRFs to shock computed under different weighting schemes



Notes: 68% and 90% confidence bands.

## 7.4. Learning period

In the baseline specification I construct reference distributions by aggregating temperatures observed in the five years prior to the shock. Here, I assume that learning takes longer and construct those distributions over 10-year rolling windows. The shock, plotted in Figure 14, is very similar to the baseline version (95% correlation).



Figure 14: Shock computed with 10-year reference distribution

#### 7.5. Monthly temperature shocks

Throughout the paper, temperature shocks have been constructed at quarterly frequency. I here propose a monthly version of the shock that, following the same logic of the baseline, starts from computing monthly surprises at county level. Surprising hot and cold days within each month are assessed with respect to a rolling, reference distribution computed by pooling daily average temperatures recorded for the same month of the year during the past 5 years; county-level surprises are then aggregated using annual population weights to get a US-wide monthly time series. To estimate the economic effect of temperatures at the higher, monthly frequency, a local projection estimate is carried out using this monthly version of the shock. Regarding the response variables, given the unavailability of GDP at such frequency, an indicator of industrial production is used to evaluate the output effect on the industry sector; monthly Consumer Price Index, 3-month rate and 10-year yield are employed as additional response variables. Results, show in Figure 15, suggest that the direction, size and timing of the response is in line with the baseline. The response of industrial production is negative as that of GDP but somewhat lagged, maybe due to the positive impact on the utility sector found in Figure 7.

### 8. Other estimates

In this Section I explore the economic effects of temperature shocks along other dimensions, including impacts at local level, a focus on the central bank reaction, the different



Figure 15: IRFs to shock computed at monthly frequency

Notes: 68% and 90% confidence bands.

implications of heat vs cold shocks and of temperature shocks over time.

#### 8.1. Local effects of temperatures

According to the shock formulation in Section 4.2, county-level temperature data are used to estimate US-wide economic effects. One interesting question that arises is how this aggregate effect is made up, i.e. how temperatures are able to affect the economy at local level. I take up this question by estimating the impacts of temperatures on local output. As county-level GDP is only available since 2001, I resort to an estimate of the effects of temperatures at state level and annual frequency, as it is also done in Colacito et al. (2019). For this purpose, I first sum the quarterly county-level surprises within each year to obtain yearly county-level series; then, I compute annual state-level shocks by weighting county-level surprises with population shares within the state<sup>23</sup> As the series of real Gross State Product (GSP) is consistently available for each state only since 1998, the estimation sample is here restricted to the 1998-2019 period.<sup>24</sup> Figure 16 shows the response to a one-standard deviation annual state-level shock obtained with local projections. In the left panel, pointwise median responses are shown together with 10-90 percentiles for horizons up to 3 years ahead. While the response of GSP is quite heterogeneous on impact in this sample, it becomes predominantly negative between one and two years after the shock, in line with what is found in the baseline exercise at national level. Indeed, the right panel shows that the vast majority of the US states experience a slowdown in real GSP two years after the shock; the size of the response is in line with the baseline, albeit on average smaller (if scaled by a factor of four), as expected for the most recent period.

Figure 16: Annual state-level IRFs (1998-2019): real GSP



Notes: Left panel: pointwise median state-level IRFs (solid line) and 10-90 percentiles (dashed lines) from a one-std deviation yearly state-level shock. Right panel: IRFs at 2y horizon, by US states.

<sup>23</sup>In formula, annual county surprises and state-level temperature shocks are

county\_surprise<sup>*i*</sup><sub>*y*</sub> = 
$$\sum_{j=1}^{4} \left[ \text{county_surprise}^{i}_{j,y} \right]$$
  $i \in k$  (8)

state\_shock<sub>y</sub><sup>p</sup> = 
$$\sum_{i=1}^{S_p} \left( \text{county}\_\text{surprise}_y^i \times w\_\text{state}_{y-1}^i \right)$$
 (9)

where j=q1,...,q4, w\_state is the state-level population weight of county *i* and  $S_p$  is the number counties in state *p*.

<sup>24</sup>See the Cautionary Note About Annual GDP by State Discontinuity on the Bureau of Economic Analysis's website.

#### 8.2. Central bank reaction

The impact of temperature shocks on the short-term rate observed in Figure 4 suggests that the Fed does react by lowering policy rates. A natural question that arises is whether temperature shocks are also able to affect the central bank's view on the economy, and whether shocks stimulate some specific discussion within the Federal Open Market Committee (FOMC).

To have insights on on the former, I re-run local projections on Greenbook Forecasts for GDP and CPI for quarter t+2, to see whether the Fed is able to foresee a persistent effect of temperatures beyond impact effects embedded in economic nowcasts. Greenbook forecasts for GDP and CPI are produced for the FOMC prior to each meeting and, at the time of the estimates, were jointly available to the public between 1980 and 2015. Figure 17 compares the response of GDP and CPI in the same sample period (first column) with the effect on GDP and CPI forecasts for quarter t+2 (GF:GDP<sub>*t*+2</sub> and GF:CPI<sub>*t*+2</sub>). While output effects are not expected to persist when the shock hits – at time 0, confidence bands of the t+2 GDP forecast cross the zero line – GDP forecasts are then adjusted to include some, still persisting temperature effect. Differently, the predicted response of the CPI goes to the opposite direction with respect to where it eventually goes. All in all, while the central bank displays a timely reaction by lowering short rates, the identification of the current shock as a persisting drag to aggregate demand seems to be a much harder task.

Do temperature shocks raise the attention of the FOMC regarding weather variations? I analyze the transcripts of all FOMC meetings historically available (i.e., from 1976 to 2015) and define three sets of weather-related terms: a wording related to the climate change phenomenon (Climate change wording); a wording related to temperatures and strictly related phenomena (Temperatures wording); a wording related to climate-related non-temperature events (Natural disasters wording).<sup>25</sup> Figure 18 displays the word count over time for each set of words. The first panel reveals that the FOMC has explicitely mentioned the climate change phenomenon very rarely (only 7 times since 1976), but temperatures and, to a lesser extent, natural disasters, quite frequently. Overall, evidence suggests that during the most recent phase of the global warming era, climate change has rarely been a hot topic in official central bank's conversations, at least before 2016 (temperatures had mentioned, on average, just 4 times per quarter).<sup>26</sup> Using the temperature and natural disaster word counts as dependent variables in local projections, it turns out that the FOMC slightly increases mention of temperatures after shocks, suggesting some moderate attention devoted to weather variations by the Committee (Figure 19).



Figure 17: Effects on Fed's Greenbook Forecasts

Notes:  $GF:X_t$  is Greenbook Forecast for time t, variable X.

 $^{25}$ The three sets of words are the following:

- Climate change: climate change, climate crisis, climate emergency, climate breakdown, global warming, global heating, carbon emissions, greenhouse gas emissions
- Temperatures: extreme heat, heat wave, temperature, hot days, weather, cold days, drought, wildfires, heat stroke, sunstroke
- Climate-related non-temperature events: flood, landslide, tornado, hurricane, dustnado, snowfall, snow, rainfall, precipitation

<sup>26</sup>Nonetheless, this attitude could have changed since then, as climate change debate increased strongly in popularity.



Figure 18: Climate mentions by FOMC members

Figure 19: Effect on Fed's climate-related wording



Notes: 68% and 90% confidence bands.

## 8.3. Heat vs cold shocks

The baseline temperature surprise shock is constructed by summing left and right tail surprises in each quarter. In this section I dissect the economic effects of these two components by including them separately in local projection estimates. According to Equations 4 and 5, left- and right-tail shocks, which I call cold and heat surprise shocks, can be retrieved as

US\_shock<sup>b</sup><sub>t,y</sub> = 
$$\sum_{i=1}^{K} \left( \text{county}\_\text{surprise}_{t,y}^{i,b} \times w_{y-1}^{i} \right) \qquad b = \{\text{cold}, \text{heat}\}$$
(10)

where

$$\begin{aligned} \text{county\_surprise}_{t,y}^{i,\text{cold}} &= \sum_{j=1}^{n_t^i} \ \mathbf{I}(T_{d,t,y}^i < \mathbf{lt}_{t,y}^i) - N_{t,y} \times 0.1 \times 0.2 \\ \text{county\_surprise}_{t,y}^{i,\text{heat}} &= \sum_{j=1}^{n_t^i} \ \mathbf{I}(T_{d,t,y}^i > \mathbf{ut}_{t,y}^i) - N_{t,y} \times 0.1 \times 0.2 \end{aligned}$$

I then employ those variables into linear local projections as

$$y_{t+s} = \alpha_s + \gamma_s \text{ US\_shock}_t^{\text{cold}} + \delta_s \text{ US\_shock}_t^{\text{heat}} + \psi_s(L)\mathbf{X}_{t-1} + u_{t+s} \qquad s = 0, 1, 2, \dots, H$$
(11)

where  $\gamma_s$  and  $\delta_s$  are impulse response functions to cold and heat shocks, respectively, and **X** is the set of control variables. Figures 20 and 21 show the response to the two shocks of all the variables considered in the baseline estimates. The economic impact of heat and cold shocks looks overall quite similar, although the response to cold shocks is somewhat larger (especially for GDP, consumption and investment) and more significant.



Figure 20: IRFs to heat shock

Notes: 68% and 90% confidence bands.

#### 8.4. Impacts by season

How does the impact of heat and cold shocks vary by season? For this purpose, I estimate the following variant of Equation 11 :

$$y_{t+s} = \alpha_s + \sum_{j=1}^{4} \left( \gamma_s^j \text{ US\_shock}_t^{\text{cold}} \times D_t^j + \delta_s^j \text{ US\_shock}_t^{\text{heat}} \times D_t^j \right) + \psi_s(L) \mathbf{X}_{t-1} + u_{t+s}$$
(12)

where s = 0, 1, 2, ..., H.  $D_t^j$  with  $j = \{1, ..., 4\}$  are four dummies that equal 1 if current quarter is quarter j of the year, 0 otherwise. As I interpret quarters as calendar seasons,

 $<sup>^{26}</sup>$ The set of controls is the same I employed in the baseline estimate. Four lags of the shocks are substituted here with four lags of both heat and cold shocks.



Figure 21: IRFs to cold shocks

Notes: 68% and 90% confidence bands.

 $\gamma_s^j$  and  $\delta_s^j$  represent impulse response functions to cold and heat shocks in winter (q1), spring (q2), summer (q3) and fall (q4). Figure 22 and Figure 23 represent IRFs for heat and cold shocks by season, which are displayed for GDP, CPI and short rate only to save space.<sup>27</sup> Overall, the two figures show that the impact of heat and cold shocks are generally more pronounced in summer than in other seasons. Regarding heat shocks, summer shocks have a stronger impact on GDP and the short rate than shocks in other seasons; the sign of the response of CPI vary substantially across seasons, appearing to be positive in winter and fall. Instead, cold temperature shocks have a detrimental economic effect not only in summer, but also in winter, while cold shocks in spring and fall seem to play a minor role.

 $<sup>^{27}\</sup>mathrm{IRFs}$  to heat and cold shocks by season of all the other variables in the baseline are available upon request.



Figure 22: IRFs to heat shocks by season

Figure 23: IRFs to cold shocks by season



## 8.5. Impact over time

The impacts found in the baseline estimates are the average effects of the shock throughout the sample period. However, as climate manifestations are the more and more visible and knowledge about them has become more widespread over time, the size of the economic impacts may have also changed. To gauge such potential time variation, I repeat local projections on sub-periods, focusing on the effects on GDP, CPI and the 3-month rate only. I set a trailing sample of 144 quarters (36 years) and perform new estimates by rolling it by one quarter at a time. In this way, I get 37 sets of estimates, which begin from the sample period 1975q1-2010q4 and end with the sample 1984q1-2019q4. To summarize the impulse responses obtained at each iteration, I display for both variables the IRFs at the horizon displaying the largest negative impact (trough) at each iteration.





Figure 24 shows the results for GDP, CPI and the short rate, where pointwise median responses are displayed together with their 68% and 90% (pointwise) confidence bands. The estimates suggest two interesting facts. First, the elasticities of GDP, CPI and the short rate to the shock have slightly reduced in size over time: for example, the CPI passed from around -0.4% to about -0.10% This reduced impact may reflect the fact that the increasing adoption of heating and cooling systems in houses and firm establishments and, more generally, the adaptation measures put in place to safeguard from temperature oscillations have made agents more resilient to shocks than they were 50 years ago. However, temperature shocks continue to have a significantly negative impact on the US economy, inducing a concomitant fall in GDP, CPI and the short-term interest rate also in the most recent part of the sample.

## 9. Conclusions

The relation between climate and the economy is typically studied relating realized temperature changes to GDP. Leveraging on the insights of the empirical macroeconomic literature, I focus instead on the economic impact of temperature surprises, i.e. unexpected fluctuations in the occurrence of exceptionally hot or cold days obtained from high-frequency, granular data for the USA. I isolate unexpected temperature shocks based on the incidence of relatively high and low temperatures in excess what agents expect in each location and season, and then aggregate local shocks to the country level using population weights. Estimates made with local projections show that temperature surprise shocks in the United States have significantly damaged the economy in the most recent phase of the global warming era, and that the Federal Reserve has been reacting to the deteriorating environment by lowering short-term rates. All in all, results show that temperatures are an autonomous source of macroeconomic variation, adding another piece of evidence in the debate on the needed policy response to climate change. The shock constructed in this paper can be replicated for other countries and at wider or narrower geographic dimension, and can serve as reference to build other weather-related shocks.

## References

- Acevedo, S., Mrkaic, M., Novta, N., Pugacheva, E., Topalova, P., 2020. The effects of weather shocks on economic activity: What are the channels of impact? Journal of Macroeconomics 65, 103–207.
- Addoum, J.M., Ng, D.T., Ortiz-Bobea, A., 2020. Temperature Shocks and Establishment Sales. The Review of Financial Studies 33, 1331–1366.
- Addoum, J.M., Ng, D.T., Ortiz-Bobea, A., 2021. Temperature Shocks and Industry Earnings News. Mimeo, available at SSRN.
- Alessandri, P., Mumtaz, H., 2021. The macroeconomic cost of climate volatility. Papers 2108.01617. arXiv.org.
- American Academy of Actuaries, Canadian Institute of Actuaries, Casualty Actuarial Society and Society of Actuaries, 2020. Actuaries climate index .
- Baker, J., Cameron, M., 1996. The Effects of the Service Environment on Affect and Consumer Perception of Waiting Time: An Integrative Review and Research Propositions. Journal of the Academy of Marketing Science 24, 338–349.
- Barreca, A., Clay, K., Deschenes, O., Greenstone, M., Shapiro, J.S., 2016. Adapting to climate change: The remarkable decline in the us temperature-mortality relationship over the twentieth century. Journal of Political Economy 124, 105–159.

- Berg, K.A., Curtis, C.C., Mark, N.C., 2021. Gdp and temperature: Evidence on crosscountry response heterogeneity. mimeo .
- Bortolan, L., Dey, A., Taschini, L., 2022. Nailing down volatile temperatures: Examining their effects on asset prices. Available at SSRN: https://ssrn.com/abstract=4023755.
- Burke, M., Davis, W., Diffenbaugh, N., 2018. Large potential reduction in economic damages under un mitigation targets. Nature 557, 549–553.
- Burke, M., Hsiang, S., Miguel, E., 2015. Global non-linear effect of temperature on economic production. Nature 527, 235–239.
- Cachon, G.P., Gallino, S., Olivares, M., 2012. Severe weather and automobile assembly productivity. Columbia Business School Research Paper No. 12/37, Available at SSRN.
- Carleton, T., Jina, A., Delgado, M., Greenstone, M., Houser, T., Hsiang, S., Hultgren, A., Kopp, R.E., McCusker, K.E., Nath, I., Rising, J., Rode, A., Seo, H.K., Viaene, A., Yuan, J., Zhang, A.T., 2022. Valuing the Global Mortality Consequences of Climate Change Accounting for Adaptation Costs and Benefits\*. The Quarterly Journal of Economics 137, 2037–2105.
- Carleton, T.A., Hsiang, S.M., 2016. Social and economic impacts of climate. Science 353, aad9837.
- Cascarano, M., Natoli, F., Petrella, A., 2022. Entry, exit and market structure in a changing climate. MPRA Paper No. 112868, University Library of Munich, Germany.
- Cashin, P., Mohaddes, K., Raissi, M., 2017. Fair weather or foul? the macroeconomic effects of el nino. Journal of International Economics 106, 37–54.
- Castro-Vincenzi, J., 2022. Climate hazards and resilience in the global car industry. Job Market Paper .
- Choi, D., Gao, Z., Jiang, W., 2020. Attention to Global Warming. The Review of Financial Studies 33, 1112–1145.

- Ciccarelli, M., Marotta, F., 2021. Demand or supply? An empirical exploration of the effects of climate change on the macroeconomy. Working Paper Series 2608. European Central Bank.
- Colacito, R., Hoffmann, B., Phan, T., 2019. Temperature and growth: A panel analysis of the united states. Journal of Money, Credit and Banking 51, 313–368.
- Custódio, C., Ferreira, M.A., Garcia-Appendini, E., Lam, A., 2021. Economic Impact of Climate Change. Mimeo, available at SSRN.
- Dell, M., Jones, B., Olken, B., 2012. Temperature shocks and economic growth: Evidence from the last half century. American Economic Journal: Macroeconomics 4, 66–95.
- Dell, M., Jones, B.F., Olken, B.A., 2014. What Do We Learn from the Weather? The New Climate-Economy Literature. Journal of Economic Literature 52, 740–98.
- Deryugina, T., 2013. How do people update? The effects of local weather fluctuations on beliefs about global warming. Climatic Change 118, 397–416.
- Deryugina, T., Hsiang, S.M., 2014. Does the Environment Still Matter? Daily Temperature and Income in the United States. NBER Working Papers 20750. National Bureau of Economic Research, Inc.
- Deschenes, O., 2014. Temperature, human health, and adaptation: A review of the empirical literature. Energy Economics 46, 606–619.
- Deschênes, O., Greenstone, M., 2007. The economic impacts of climate change: Evidence from agricultural output and random fluctuations in weather. American Economic Review 97, 354–385.
- Diebold, F., Rudebusch, G., 2022. On the Evolution of US Temperature Dynamics. Chudik, A., Hsiao, C. and Timmermann, A. (Ed.) Essays in Honor of M. Hashem Pesaran: Prediction and Macro Modeling - Advances in Econometrics 43.
- Donadelli, M., Jüppner, M., Paradiso, A., Schlag, C., 2021. Computing macro-effects and welfare costs of temperature volatility: A structural approach. Computational Economics 58, 347–394.

- Donadelli, M., Jüppner, M., Riedel, M., Schlag, C., 2017. Temperature shocks and welfare costs. Journal of Economic Dynamics and Control 82, 331–355.
- Faccia, D., Parker, M., Stracca, L., 2021. Feeling the heat: extreme temperatures and price stability. ECB Working Paper Series .
- Fisher, A.C., Hanemann, W.M., Roberts, M.J., Schlenker, W., 2012. The economic impacts of climate change: Evidence from agricultural output and random fluctuations in weather: Comment. American Economic Review 102, 3749–60.
- Forni, M., Gambetti, L., 2016. Government spending shocks in open economy VARs. Journal of International Economics 99, 68–84.
- Gallic, E., Vermandel, G., 2020. Weather shocks. European Economic Review 124, 103409.
- Gertler, M., Karadi, P., 2015. Monetary policy surprises, credit costs, and economic activity. American Economic Journal: Macroeconomics 7, 44–76.
- Graff Zivin, J., Neidell, M., 2014. Temperature and the allocation of time: Implications for climate change. Journal of Labor Economics 32, 1 26.
- Griffit, W., Veitch, R., 1971. Hot and crowded: Influence of population density and temperature on interpersonal affective behavior. Journal of Personality and Social Psychology 17(1), 92–98.
- Gurkaynak, R.S., Sack, B., Swanson, E., 2005. Do Actions Speak Louder Than Words? The Response of Asset Prices to Monetary Policy Actions and Statements. International Journal of Central Banking 1.
- Hansen, L.P., 2021. Central banking challenges posed by uncertain climate change and natural disasters. Journal of Monetary Economics .
- Hong, H., Wang, N., Yang, J., 2020. Mitigating Disaster Risks in the Age of Climate Change. NBER Working Papers 27066. National Bureau of Economic Research, Inc.
- Hsiang, S., Kopp, R., Jina, A., Rising, J., Delgado, M., Mohan, S., Rasmussen, D.J., Muir-Wood, R., Wilson, P., Oppenheimer, M., Larsen, K., Houser, T., 2017. Estimating economic damage from climate change in the united states. Science 356, 1362–1369.

- Jin, Z., Li, F.W., Lin, Y., Zhang, Z., 2021. Do firms adapt to rising temperatures? evidence from establishment-level data. Available at SSRN. .
- Jorda, 2005. Estimation and Inference of Impulse Responses by Local Projections. American Economic Review 95, 161–182.
- Kala, N., 2019. Learning, adaptation, and climate uncertainty: Evidence from indian agriculture. MIT Center for Energy and Environmental Policy Research Working 50.
- Kelly, D., Kolstad, C., Mitchell, G.T., 2005. Adjustment costs from environmental change. Journal of Environmental Economics and Management 50, 468–495.
- Kiley, M.T., 2021. Growth at Risk From Climate Change. Finance and Economics Discussion Series 2021-054. Board of Governors of the Federal Reserve System (U.S.).
- Kim, H.S., Matthes, C., Phan, T., 2021. Extreme Weather and the Macroeconomy. Working Paper 21-14. Federal Reserve Bank of Richmond.
- Kolstad, C.D., Moore, F.C., 2020. Estimating the economic impacts of climate change using weather observations. Review of Environmental Economics and Policy 14, 1–24.
- Kotz, M., Wenz, L., Stechemesser, A., Kalkuhl, M., Levermann, A., 2021. Day-to-day temperature variability reduces economic growth. Nature Climate Change 11, 319–325.
- Lemoine, D., 2021. Estimating the consequences of climate change from variation in weather. CEPR Discussion Paper 16194.
- Makkonen, A., Vallström, D., Uddin, G.S., Rahman, M.L., Haddad, M.F.C., 2021. The effect of temperature anomaly and macroeconomic fundamentals on agricultural commodity futures returns. Energy Economics 100, 105377.
- Malmendier, U., Nagel, S., 2015. Learning from Inflation Experiences. The Quarterly Journal of Economics 131, 53–87.
- McFarland, J., Z.Y.C.L.e.a., 2015. Impacts of rising air temperatures and emissions mitigation on electricity demand and supply in the United States: a multi-model comparison. Technical Report.

- Miranda-Agrippino, S., Ricco, G., 2021. The transmission of monetary policy shocks. American Economic Journal: Macroeconomics 13, 74–107.
- Moore, F.C., 2017. Learning, adaptation, and weather in a changing climate. Climate Change Economics 08, 1750010.
- Morss, R.E., Lazo, J.K., Demuth, J.L., 2010. Examining the use of weather forecasts in decision scenarios: results from a us survey with implications for uncertainty communication. Meteorological Applications 17, 149–162.
- Moscona, J., Sastry, K.A., 2022. Does Directed Innovation Mitigate Climate Damage? Evidence from U.S. Agriculture\*. The Quarterly Journal of Economics .
- Mukherjee, K., Ouattara, B., 2021. Climate and monetary policy: do temperature shocks lead to inflationary pressures? Climatic change 167.
- Nakamura, E., Steinsson, J., 2018a. High-Frequency Identification of Monetary Non-Neutrality: The Information Effect. The Quarterly Journal of Economics 133, 1283– 1330.
- Nakamura, E., Steinsson, J., 2018b. Identification in macroeconomics. Journal of Economic Perspectives 32, 59–86.
- National Weather Service, 1989. The december 1989 cold wave.
- Pankratz, N., Schiller, C., 2021. Climate Change and Adaptation in Global Supply-Chain Networks, in: Proceedings of Paris December 2019 Finance Meeting EUROFIDAI-ESSEC, European Corporate Governance Institute–Finance Working Paper.
- Ramey, V., 2016. Chapter 2 macroeconomic shocks and their propagation, Elsevier. volume 2 of Handbook of Macroeconomics, pp. 71 162.
- Roth Tran, B., 2022. Sellin' in the Rain: Weather, Climate, and Retail Sales. Technical Report. Federal Reserve Bank of San Francisco Working Paper.
- Schlenker, W., Roberts, M.J., 2009. Nonlinear temperature effects indicate severe damages to u.s. crop yields under climate change. Proceedings of the National Academy of Sciences 106, 15594–15598.

- Somanathan, E., Somanathan, R., Sudarshan, A., Tewari, M., 2021. The Impact of Temperature on Productivity and Labor Supply: Evidence from Indian Manufacturing. Journal of Political Economy 129, 1797–1827.
- Starr-McCluer, M., 2000. The effects of weather on retail sales. Technical Report.
- Sterba, J.P., 1977. 1977, a year of weather extremes, causing concern among experts.

#### Appendix A. Diagnostics on the shocks

In order to check that my US-wide shock series has desirable properties to be used in macroeconomic analysis, I perform the Ljung-Box Q-test for serial correlation. I repeat the test 6 times including a different number of lags each time: 1,4,8,12,16 and 20 quarters. All tests fail to reject the null hypothesis of no residual autocorrelation. The autocorrelation function is displayed in Figure Appendix A.1, showing that evidence for serial correlation is limited at best.





## Appendix B. Forecasting surprisingly hot and cold days

Expectations on the temperature distribution at quarterly frequency are fundamentally backward-looking. Assuming agents being able to quickly adjust their plans in exceptionally hot or cold days, can publicly available weather forecasts really help? To have some insights on this question, I here conduct an illustrative exercise to investigate whether days that are labelled as surprisingly hot or cold according to my technique are also more difficult to predict. I use data from the Model Output Statistics (MOS) database of the National Weather Service, collected from the Iowa State Mesonet archive and available since year 2000. I focus on Washington DC county and construct predictions of one- and two-day ahead average daily temperatures at daily frequency.<sup>28</sup> Using

<sup>&</sup>lt;sup>28</sup>DC county is chosen because temperatures are recorded through one weather station only, simplifying the calculations. From that station, forecasts are made six times per day and relate to temperatures for the same day and up to 3 days ahead, at the 3-hour frequency. To construct predictions, I retain the

realized county temperatures, I construct daily ex-post forecast errors as

$$\operatorname{fe}_t^{\ j} = f_{t-j}^{\ j} - \operatorname{temp}_t$$

where  $f_{t-j}^{\ j}$  are forecasts made at time t-j with horizon j, with j = 1, 2 days and  $temp_t$  are the realized average daily temperatures used to construct my temperature shock. To compare forecasts with surprises, I compute a dummy variable that equals one if the day is labeled as surprisingly hot or cold under my procedure, and 0 otherwise. I then estimate a probit regression model to test if high forecast errors (in absolute values) are correlated with a higher probability of a day being labeled as surprising. In formulas

Prob<sub>t</sub> (surprising day<sub>t</sub> = 1) = 
$$F\left[\alpha + \beta_j |\text{fe}_t^j|\right]$$
  $j = 1, 2$ 

The estimates are carried out in the sample going from Jul 1, 2000 to December 31, 2019. Results, available upon request, show that the probability of a day being surprising is positively and significantly correlated with the size of forecasts errors, confirming that days with temperatures going beyond expectations are also more difficult to predict than those with milder temperatures.

### Appendix C. Narrative on the shock

1977q1: As the New York Times described, 1977 was "a year of weather extremes", which spurred new climate research in the United States and worldwide (Sterba, 1977). Analyses made at that time by the National Weather Service reported that winter in 1977, especially in the eastern part of the country, was one of the coldest in the 20th century, with temperatures close to the record lows of 1917-18.

1989q4: The fourth quarter of 1989 also stood apart for its exceptionally low temperatures. Surges of Arctic air in December was, according to the National Weather Service, "a historic event, with many locations establishing monthly or all-time record lows" (National Weather Service, 1989). While the fall in temperatures was strongest in the northwest, the southeastern coast was also particularly damaged, with losses caused by damaged crops in Florida and broken water pipes in Texas causing failures at manufacturing plants. In that area, the 1989 cold wave also generated the largest snowstorm in

last forecasts made during the day (8 pm DC time).

recent history.

2000q4: The fourth quarter of year 2000 saw temperatures passing from one extreme to the other. The late summer period was characterized by a prolonged heat wave which, particularly, in southern states, resulted in record high temperatures. Then, after a warm October, temperatures plunged in November – the second coldest since records began in 1895, according to NOAA's National Climate Report. This turnaround in temperatures caused power outages and triggered other weather-related extreme events, such as tornadoes.

2003q1: The fourth peak in the series is that of winter of 2003. That period will be mainly remembered in northern states such as Minnesota and Michigan. While it was not among the coldest on record, it stood out for its persisting cold, which started in December and went through mid-March. In particular, Michigan recorded the second longest continuous streak of very cold temperatures (76% of days in January and February) on record, just one notch below the exceptionally cold winter of 1977.

# Appendix D. Data sources

Variable	Source	Frequency	Sample
Temperature shock			
gridded average daily temperatures	Northeast Begional Climate Center	daily	Jan 1 1970 - Dec 31 2019
population (by county)	Census Bureau	vearly	1969 - 2018
employment (by county)	Census Bureau	vearly	1969 - 2018
personal income (by county)	Census Bureau	vearly	1969 - 2018
land extension (by county)	Census Bureau	yearly	1969 - 2018
Real GDP (by county) (CAGDP9)	Bureau of Economic Aalysis (BEA)	yearly	2001
Response variables			
Real GDP (USGDPD)	Refinitiv	quarterly	1975q1-2019q4
Real Personal Consumption Expenditures (USCNPER.D)	Refinitiv	quarterly	1975q1-2019q4
Real gross private domestic investment (USGDPRIND)	Refinitiv	quarterly	1975q1-2019q4
Consumer Price Index (USCONPRCE)	Refinitiv	quarterly	1975q1-2019q4
Population (USPOPTOTP)	Refinitiv	quarterly	1975q1-2019q4
Unemployment level (USUNPTOTO)	Refinitiv	quarterly	1975q1-2019q4
Unemployment rate (USUN%TOTQ)	Refinitiv	quarterly	1975q1-2019q4
3-month rate (TB3MS)	Fred	quarterly	1975q1-2019q4
1-year Treasury yield (GS2)	Fred	quarterly	1975q1-2019q4
10-year Treasury yield (GS10)	Fred	quarterly	1975q1-2019q4
PCE: Energy goods and services (DNRGRC1Q027SBEA)	Fred	quarterly	1975q1-2019q4
CPI: Energy (CPIENGSL)	Fred	quarterly	1975q1-2019q4
CPI: Food and Beverages (CPIFABSL)	Fred	quarterly	1975q1-2019q4
CPI: All Items Less Food and Energy (CPILFESL)	Fred	quarterly	1975q1-2019q4
GDP deflator (GDPDEF)	Fred	quarterly	1975q1-2019q4
Real Gross Value Added - Business - Farm	BEA NIPA Table 1.3.3	quarterly	1975q1-2019q4
Real Gross Value Added - Business - Non-Farm	BEA NIPA Table 1.3.3	quarterly	1975q1-2019q4
Real Personal Consumption Expenditures by Type of Product	BEA NIPA Table 2.3.3	quarterly	1975q1-2019q4
Real Private Fixed Investment by Type	BEA NIPA Table 5.3.3	quarterly	1975q1-2019q4
Real Gross State Product	BEA Regional Data	yearly	1998-2019
Employment level	Bureau of Labor Statistics	quarterly	1975q1-2019q4
Hours worked, business sector	Bureau of Labor Statistics	quarterly	1975q1-2019q4
Labor productivity, business sector	Bureau of Labor Statistics	quarterly	1975q1-2019q4
Industrial production - manufacturing	Federal Reserve Board G.17	quarterly	1975q1-2019q4
Industrial production - utility	Federal Reserve Board G.17	quarterly	1975q1-2019q4
GDP forecasts $(t+2)$	Greenbook Forecasts	quarterly	1980q1-2015q4
CPI forecasts (t+2)	Greenbook Forecasts	quarterly	1980q1-2015q4
FOMC word count			
Transcripts of FOMC meetings	federalserve.gov	meeting calendar	1976 - 2015
Weather forecasts			
1-day ahead temperature forecasts (at 8 pm DC time) $$	Iowa State Mesonet archive	daily	Jul 1,2000 - Dec 31,2019
2-day ahead temperature forecasts (at 8 pm DC time)	Iowa State Mesonet archive	daily	Jul 1,2000 - Dec 31,2019

## Table Appendix D.1: Data summary

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 1383 Bank lending to small firms: metamorphosis of a financing model, by Paolo Finaldi Russo, Valentina Nigro and Sabrina Pastorelli (October 2022).
- N. 1384 Macroeconomic effects of growth-enhancing measures in the euro area, by Alessandro Cantelmo, Alessandro Notarpietro and Massimiliano Pisani (October 2022).
- N.1385 *Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel*, by Claire Giordano (October 2022).
- N. 1386 Labor market spillovers of a large plant opening. Evidence from the oil industry, by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani (October 2022).
- N. 1387 Unburdening regulation: the impact of regulatory simplification on photovoltaic adoption in Italy, by Federica Daniele, Alessandra Pasquini, Stefano Clò and Enza Maltese (October 2022).
- N.1388 *Liberalizing the opening of new pharmacies and hospitalizations*, by Andrea Cintolesi and Andrea Riganti (November 2022).
- N.1389 *Mediation Analysis Synthetic Control*, by Giovanni Mellace and Alessandra Pasquini (November 2022).
- N.1390 The effects of partial employment protection reforms: evidence from Italy, by Diego Daruich, Sabrina Di Addario and Raffaele Saggio (November 2022).
- N. 1391 *The role of majority status in close election studies*, by Matteo Alpino and Marta Crispino (November 2022).
- N. 1392 Issuing bonds during the Covid-19 pandemic: is there an ESG premium?, by Fabrizio Ferriani (November 2022).
- N. 1393 *Economic fundamentals and stock market valuation: a CAPE-based approach*, by Maria Ludovica Drudi and Federico Calogero Nucera (November 2022).
- N. 1394 *The impact of "Metro C" in Rome on the housing market*, by Federica Daniele and Elena Romito (December 2022).
- N. 1395 *Gender quotas, board diversity and spillover effects. Evidence from Italian banks,* by Silvia Del Prete, Giulio Papini and Marco Tonello (December 2022).
- N.1396 *Real-time inequalities and policies during the pandemic in the US*, by Luisa Corrado, Daniela Fantozzi and Simona Giglioli (December 2022).
- N. 1397 *The impact of sovereign tensions on bank lending: identifying the channels at work*, by Fabiana Sabatini (December 2022).
- N. 1398 Make-up strategies and exchange rate pass-through in a low-interest-rate environment, by Alessandro Cantelmo, Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (December 2022).
- N. 1399 *Measuring peer effects in parental leaves: evidence from a reform*, by Davide Dottori, Francesca Modena and Giulia Martina Tanzi (February 2023).
- N. 1400 Firms' innovation and university cooperation. New evidence from a survey of Italian firms, by Daniela Bragoli, Flavia Cortelezzi and Massimiliano Rigon (February 2023).
- N. 1401 Schools and the transmission of Sars-Cov-2: evidence from Italy, by Salvatore Lattanzio (February 2023).
- N. 1402 *Remittances in times of crisis: evidence from Italian corridors*, by Alessio Ciarlone (February 2023).
- N. 1403 *EU banks' dividend policies: main determinants and the role of capital ratios*, by Salvatore Cardillo and Jacopo Raponi (February 2023).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times,* R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014)**.
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy*, Journal of International Economics, v. 126, article 103328, **WP 1032** (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DE BLASIO G., I. DE ANGELIS and L. RIZZICA, *Lost in corruption. Evidence from EU funding to Southern Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 6, 3, pp. 355–377, WP 1180 (December 2016).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, WP 1299 (October 2020).
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, WP 1232 (July 2019).
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., *On risk factors of the stock–bond correlation,* International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, Article 103201, WP 1142 (October 2017).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- RIZZICA L. and M. TONELLO, *Persuadable perceptions: the effect of media content on beliefs about corruption*, Economic Policy, v. 35, 104, pp. 679–737, WP 1043 (November 2016).

SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).

SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, **WP 1072 (July 2016).** 

#### 2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019)**.
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno, Papers in Regional Science, v. 100, 2, pp. 327-348, WP 1253 (December 2019).
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, **WP 1133 (September 2017).**
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, *Intergenerational mobility in the very long run: Florence 1427-2011*, Review of Economic Studies, v. 88, 4, pp. 1863–1891, **WP 1060 (April 2016).**
- BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (October 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, **WP 1288 (July 2020).**
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, **WP 1222 (June 2019).**
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**

- DEL PRETE S. and M. L. STEFANI, *Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks*, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, **WP 1014 (June 2015).**
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area,* Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, WP 1101 (February 2017).
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, **WP 1277 (June 2020).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth. v. 67, 1, pp. 61-103, WP 1095 (January 2017).
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., *Demographics and the natural real interest rate: historical and projected paths for the Euro Area,* Journal of Economic Dynamics and Control, v. 132, Article 04209, WP 1306 (November 2020).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

#### 2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Covid-19 shock and fiscal-monetary policy mix in a monetary union*, Economic challenges for Europe after the pandemic, Springer Proceedings in Business and Economics, Berlin-Heidelberg, Springer, WP 1313 (December 2020).
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, v. 66, Article 100762, **WP 1326 (March 2021)**.
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility*, Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018).**
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021).**
- CONTI A. M., A. NOBILI and F. M. SIGNORETTI, *Bank capital requirement shocks: a narrative perspective,* European Economic Review, v.151, Article 104254, **WP 1199 (November 2018).**
- FAIELLA I. and A. MISTRETTA, *The net zero challenge for firms' competitiveness*, Envnmental & Resource Economics, v. 83, pp. 85-113, **WP 1259 (February 2020).**
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets: evidence from the mortgage market,* Journal of Financial Economics, v.143, 3, pp. 1209-1226, **WP 1252 (December 2019).**
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).

- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference*, Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, v. 18, 4, pp. 325-377, **WP 1171 (April 2018).**
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, WP 1235 (October 2019).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).
- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, **WP 1189 (September 2018).**
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions*, Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

#### FORTHCOMING

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of textbased indicators in forecasting Italian economic activity*, International Journal of Forecasting, WP 1321 (March 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Non-standard monetary policy measures in non-normal times,* International Finance, **WP 1251 (November 2019).**
- CIAPANNA E, S. MOCETTI and A. NOTARPIETRO, *The macroeconomic effects of structural reforms: an empirical and model-based approach*, Economic Policy, **WP 1303 (November 2022).**
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *It ain't where you're from, it's where you're at: hiring origins, firm heterogeneity, and Wages, Journal of Econometrics*, WP 1374 (June 2022).
- FERRARI A. and V. NISPI LANDI, *Whatever it takes to save the planet? Central banks and unconventional green policy*, Macroeconomic Dynamics, **WP 1320 (February 2021).**
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, **WP 1358 (February 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, WP 1336 (June 2021).
- LOBERTO M, Foreclosures and house prices, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOMBARDI M. J., M. RIGGI and E. VIVIANO, *Bargaining power and the Phillips curve: a micro-macro analysis, and wages,* Journal of the European Economic Association, WP 1302 (November 2020).
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy*, The Journal of Law and Economics, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, WP 1312 (December 2020).